Author's response to reviews

Title: A Rare Cause of Small Bowel Obstruction: Phytobezoar in a Jejunal Diverticulum.

Authors:

mohammad tayeb (drmtayeb@yahoo.com)
faiz khan (faizkhan.surg@yahoo.com)
fozia rauf (farauf@yahoo.com)
mumtaz khan (mmutazkhan@gmail.com)

Version: 3 Date: 22 April 2011

Author's response to reviews:

Cover letter
To
The JMCR Editorial Team
Subject: Making the suggested changes by reviewers in our Article
Case report: A Rare Cause of Small Bowel Obstruction: Phytobezoar in a Jejunal Diverticulum.
Authors:
mohammad tayeb (drmtayeb@yahoo.com)
faiz khan (faizkhan.surg@yahoo.com)
fozia rauf (farauf@yahoo.com)
mumtaz khan (mmutazkhan@gmail.com)
MS: 1624372875499765
The editor, associate editor and both of the reviewers (Arnaldo Andreoli and Anna Kohn) suggested a couple of changes.
1: The introduction and discussion is too long and needs to be shortened, I did that in this revised manuscript by reducing introduction part from 304 words to 232 words and discussion part from 527 words to 430 words.
2: A spelling error on page 3 line 10 "detaction" instead of "detection". I did that as well in this revised version.

I am grateful to the reviewers and editorial team for their quick response, valuable suggestions and help regarding publication of this article.

With Best Wishes
Mohammad Tayeb